Author’s response to reviews

Title: An implementation science protocol of the Women's Health CoOp in healthcare settings in Cape Town, South Africa: A stepped-wedge design

Authors:
Wendee Wechsberg (wmw@rti.org)
Jacqueline Ndirangu (jndirangu@rti.org)
Ilene Speizer (speizer@email.unc.edu)
William Zule (zule@rti.org)
Winnifred Gumula (Winnifred.Gumula@khethimpilo.or)
Courtney Peasant (cpeasant@rti.org)
Felicia Browne (fbrowne@rti.org)
Laura Dunlap (ljd@rti.org)

Version: 1 Date: 22 Aug 2017

Author’s response to reviews:

In response, this revised manuscript has, we have:

1. Attached the Figure 1.

2. Deleted the ethical approval documents and will send in a separate email along with proof of NIH funding.